ORIGINAL ARTICLE



Long-term exposure to elevated carbon dioxide does not alter activity levels of a coral reef fish in response to predator chemical cues

Josefin Sundin¹ · Mirjam Amcoff^{2,3} · Fernando Mateos-González^{3,4} · Graham D. Raby^{5,6} · Fredrik Jutfelt⁷ · Timothy D. Clark^{5,8}

Received: 28 February 2017 / Revised: 8 June 2017 / Accepted: 14 June 2017 © The Author(s) 2017. This article is an open access publication

Abstract

Levels of dissolved carbon dioxide (CO₂) projected to occur in the world's oceans in the near future have been reported to increase swimming activity and impair predator recognition in coral reef fishes. These behavioral alterations would be expected to have dramatic effects on survival and community dynamics in marine ecosystems in the future. To investigate the universality and replicability of these observations, we used juvenile spiny chromis damselfish (Acanthochromis polyacanthus) to examine the effects of long-term CO2 exposure on routine activity and the behavioral response to the chemical cues of a predator (Cephalopholis urodeta). Commencing at ~3-20 days post-hatch, juvenile damselfish were exposed to presentday CO₂ levels (~420 µatm) or to levels forecasted for the year 2100 (~1000 µatm) for 3 months of their development. Thereafter, we assessed routine activity before and after injections of seawater (sham injection, control) or

Communicated by J. Lindström

Electronic supplementary material The online version of this article (doi:10.1007/s00265-017-2337-x) contains supplementary material, which is available to authorized users.

☑ Josefin Sundin josefin@teamsundin.se; josefin.sundin@neuro.uu.se

- ¹ Department of Neuroscience, Uppsala University, Uppsala, Sweden
- ² Department of Zoology/Functional Zoomorphology, Stockholm University, Stockholm, Sweden
- ³ Section of Integrative Biology, University of Texas, Austin, TX, USA
- ⁴ Department of Collective Behaviour, Max Planck Institute for Ornithology, University of Konstanz, Konstanz, Germany

seawater-containing predator chemical cues. There was no effect of CO_2 treatment on routine activity levels before or after the injections. All fish decreased their swimming activity following the predator cue injection but not following the sham injection, regardless of CO_2 treatment. Our results corroborate findings from a growing number of studies reporting limited or no behavioral responses of fishes to elevated CO_2 .

Significance statement

Alarmingly, it has been reported that levels of dissolved carbon dioxide (CO_2) forecasted for the year 2100 cause coral reef fishes to be attracted to the chemical cues of predators. However, most studies have exposed the fish to CO_2 for very short periods before behavioral testing. Using long-term acclimation to elevated CO_2 and automated tracking software, we found that fish exposed to elevated CO_2 showed the same behavioral patterns as control fish exposed to present-day CO_2

- ⁵ Australian Institute of Marine Science, Townsville, Queensland, Australia
- ⁶ Great Lakes Institute for Environmental Research, University of Windsor, Windsor, Ontario, Canada
- ⁷ Department of Biology, Norwegian University of Science and Technology, Trondheim, Norway
- ⁸ Present address: University of Tasmania and CSIRO Agriculture and Food, Hobart, Tasmania, Australia

levels. Specifically, activity levels were the same between groups, and fish acclimated to elevated CO_2 decreased their swimming activity to the same degree as control fish when presented with cues from a predator. These findings indicate that behavioral impacts of elevated CO_2 levels are not universal in coral reef fishes.

Keywords Climate change · Ocean acidification · Pomacentridae · Olfaction · Alarm cue

Introduction

Ocean acidification, caused by increased levels of dissolved carbon dioxide (CO₂) (Caldeira and Wickett 2005; Doney et al. 2009), is expected to affect the fitness of marine organisms, especially if CO₂ levels reach those predicted for the year 2100 in a business-as-usual emission scenario (i.e., ~1000 µatm) (Ishimatsu et al. 2008; Dupont et al. 2010; Heuer and Grosell 2014). The research effort to understand how ocean acidification may affect marine organisms is important and growing rapidly. One increasingly popular topic is the potential for elevated CO₂ to impact fish behavior. Indeed, the number of papers on ocean acidification and fish behavior almost doubled between 2012 and 2015 (Clements and Hunt 2015; Browman 2016), with the majority of them reporting disturbed behaviors under elevated CO₂. Given the importance of using appropriate methods to measure seawater pH and the controversy with using pH to estimate CO₂ (Dickson 1984; Crawford and Harrison 1997; Moran 2014; Bockmon and Dickson 2015), we herein specify the measurement method used for each cited study (where direct measurements of CO_2 partial pressure (pCO₂) are indicated using µatm, and measurements of pH on the NBS and total scale are indicated using pH_{NBS} and pH_{TOTAL}, respectively).

Some studies have suggested that the reported effects of elevated CO2 on fish behavior will translate to detrimental effects on fish communities in the wild (reviewed in Heuer and Grosell 2014; Clements and Hunt 2015). For example, studies on adult and larval coral reef fishes report that CO₂ exposure increases swimming activity in observations performed in situ with the naked eye once fish are placed back into present-day reef conditions (pCO₂ 400-500 µatm) following brief exposure to forecasted CO₂ conditions (e.g., Ferrari et al. 2011a after 4 days' exposure to ~850 µatm; Munday et al. 2010 after 4 days' exposure to 850 µatm; Devine et al. 2012 after 4 days' exposure to ~960 µatm). In some instances, in which experiments were performed in laboratory settings (manually quantifying activity by counting the number of lines crossed by a fish in an aquarium), CO_2 exposure has been reported to increase activity levels two-fold (Cripps et al. 2011 4-11 days' exposure to pH_{NBS} 7.88) or even 90-fold (Munday et al. 2013 4 and 28 days' exposure to 940 μ atm) compared with animals kept at present-day levels of CO₂. There are a few exceptions, with some studies reporting no effects of elevated CO₂ on activity (e.g., Ferrari et al. 2012a 4 days' exposure to pH_{NBS} 7.89; Nowicki et al. 2012 21 days' exposure to pH_{NBS} 7.89–7.93).

The aspect of fish behavior receiving the most attention with respect to elevated CO₂ is the response to chemical cues, such as the alarm cues released from conspecifics and/or heterospecifics, or the array of chemical cues emitted from predators (reviewed in Leduc et al. 2013; Clements and Hunt 2015). Perhaps most alarmingly, several studies have reported that prey fish exposed to elevated CO₂ are attracted to chemical cues from predators rather than avoiding them (Dixson et al. 2010 11 days' exposure to pH_{NBS} 7.8; Munday et al. 2010 1-10 days' exposure to 850 µatm; Munday et al. 2012 4 days' exposure to pH_{NBS} 7.98; Nilsson et al. 2012 4 days' exposure to pH_{NBS} 7.81; Munday et al. 2013 4 and 28 days' exposure to 940 µatm). Similarly, several prey species exposed to high CO₂ have been reported to show a reduced response to chemical alarm cues from the skin of conspecifics (Ferrari et al. 2011a 4 days' exposure to ~850 µatm; Lönnstedt et al. 2013 4 days' exposure to pH_{NBS} 7.89; Welch et al. 2014 40-45 days' exposure to pH_{NBS} 7.85) and an impaired ability to learn to recognize predator chemical cues (Ferrari et al. 2012a 4 days' exposure to pH_{NBS} 7.89; Chivers et al. 2014 4 days' exposure to pH_{NBS} 7.85). CO₂-induced behavioral impairments reportedly also occur in piscivores. For example, brown dottybacks (Pseudochromis fuscus) exposed to elevated CO₂ (pH_{NBS} 7.88) for 4–11 days were reported to be less attracted to prey skin extract compared with control individuals (Cripps et al. 2011), and the elasmobranch smooth dogfish (Mustelus canis) was reported to avoid squid chemical cues after a 5-day exposure to pH_{NBS} 7.69 (Dixson et al. 2015).

Given the need to make predictions about the ecological effects of ocean acidification, it is a priority to clarify the behavioral effects in fishes acclimated to elevated CO₂ (Browman 2016). To help elucidate the effects of long-term exposure to CO₂ levels predicted for the year 2100 (i.e., ~1000 µatm; Caldeira and Wickett 2005; Doney et al. 2009), we exposed a coral reef damselfish, the spiny chromis (Acanthochromis polyacanthus), to elevated CO_2 for >80 days, before assessing routine activity levels as well as the behavioral response to predator chemical cues. Activity was measured using automated video-tracking software before and after the injection of chemical cues from the flagtail grouper (Cephalopholis urodeta), a natural predator of damselfishes on the Great Barrier Reef, Australia (Holbrook and Schmitt 2003). Given the reports of elevated CO₂ increasing activity in coral reef fishes, with implications for higher predation rates (Munday et al. 2010 1-10 days' exposure to 850 μatm; Cripps et al. 2011 4–11 days' exposure to pH_{NBS} 7.88; Devine et al. 2012 4 days' exposure to ~960 µatm;

Munday et al. 2013 4 and 28 days' exposure to 940 µatm), we predicted that we would similarly detect increased activity in the CO₂-acclimated fish used in this study. Furthermore, in a previous experiment that investigated transgenerational effects of CO₂ in A. polvacanthus, offspring reared in elevated CO₂ (at pH_{NBS} 7.85 for 40-45 days) were reported to be strongly attracted to conspecific chemical alarm cues, whereas individuals in control water strongly avoided them (Welch et al. 2014). Based on that study and the current view in the field that the behavioral effects of CO₂ are strong and highly replicable (reviewed in Clements and Hunt 2015), we predicted that the typical anti-predator behavioral response to chemical cues (a decrease in activity; see Kelley 2008 and references within) would be impaired or reversed. Thus, we predicted that activity levels would be maintained or even increase upon exposure to predator chemical cues in fish acclimated to high CO₂.

Methods

Animals and CO₂ exposure

Experiments were performed at the Australian Institute of Marine Science (AIMS; Townsville, Australia) between May and July 2015. Juvenile spiny chromis (mean \pm SD initial $N = 753, 0.019 \pm 0.015$ g initial wet weight, 10.6 ± 2.3 mm initial total length (TL), age \sim 3–20 days post-hatching) were obtained from the Reef HQ Aquarium in Townsville, Australia. The Reef HQ Aquarium is a large public aquarium, which resembles a natural coral reef environment and requires all small animals to remain vigilant to ensure they are not preyed upon. Clutches of fish were caught across a spatial and temporal scale to ensure they were from at least four breeding pairs. We chose this model species on the basis that it has been reported to suffer dramatic behavioral impairments in response to elevated CO₂, and it apparently lacks the capacity to reverse these impairments even following transgenerational acclimation (Welch et al. 2014). Notably, previous studies have reported that responses to elevated CO₂ in captive-bred fishes can be as strong as those of wildcaught fishes (e.g., Munday et al. 2013; Green and Jutfelt 2014; Pimentel et al. 2014; Ou et al. 2015; Tix et al. 2017), including in experiments where the fish had been captive-bred for many generations (Rossi et al. 2015; Pistevos et al. 2017).

Fish were transported in aerated seawater to AIMS where they were placed in twelve 25-L tanks with flow-through seawater (~3.5 L min⁻¹) from four independent 200-L sumps (three tanks per sump). After at least 24 h to recover from transport, the partial pressure of CO₂ (*p*CO₂) of half of the tanks (*N* = 6) was gradually increased to 1012 ± 137 µatm (mean ± SD) over 24 h using a CO₂ dosing system (pH stat Computers, Aqua Medic, Bissendorf, Germany) connected to solenoid valves regulating administration of 100% CO₂ gas into two of the partial-recirculation sump systems. The remaining tanks (N = 6) were kept at ambient pCO₂ levels $(424 \pm 13 \mu atm)$. Fresh seawater was flushed through each of the four sumps at $\sim 4-7$ L min⁻¹. Three air stones in each sump ensured that the water remained well-mixed and maintained dissolved oxygen at >90% air saturation. The pCO_2 levels of the holding tanks were checked every 1-4 days using a LI-820 CO₂ Gas Analyzer (LI-COR®, Lincoln, Nebraska, USA) (Table 1). Methods for measuring and manipulating pCO_2 thus followed best practice guidelines (Riebesell et al. 2010; Moran 2014; Cornwall and Hurd 2016). Fish were exposed to natural water temperatures for the region (quantified using thermal data-loggers sampling every 30 min; iButton, Maxim Integrated, San Jose, CA, USA). Temperature declined seasonally from 27.0 ± 0.3 °C during the first seven days of holding to 22.1 ± 0.6 °C during the seven days immediately prior to the commencement of the behavioral trials (Table 1). Total alkalinity of the water was measured in duplicate at the mid- and endpoints of the experiment (Table 1). The salinity remained at 35 ± 1 PSU at all times (regulated through the AIMS SeaSim aquarium facility). Water carbonate chemistry was calculated using the constants of Dickson (1990) and Lueker et al. (2000) in CO₂calc (Hansen, USGS, USA) (Table 1). Fish were fed ad libitum 1–2 times per day using commercial aquaculture pellets crushed to a powder and/or Artemia spp. nauplii. Individuals were given 12-18 h without food prior to their use in behavioral experiments to ensure they were in a post-absorptive state (as per Hamilton et al. 2014; Ramasamy et al. 2015). Tanks were cleaned weekly.

Cue preparation

Water containing predator chemical cues was collected from two tanks (25 L) housing flagtail groupers (C. urodeta). Each tank was supplied with water from one of the sump systems (either control or high CO2 water) and contained a small, lightly bubbling air stone to ensure that the water remained well-mixed. All effluent water from the predator tanks went straight to the waste drain rather than back to the sump, to ensure that the spiny chromis did not become habituated to predator chemical cues. The tank containing control water housed two groupers with body mass of 50.2 and 61.6 g (4.5 g fish per liter of water) and the tank containing water with elevated pCO_2 housed two groupers with body mass of 33.4 and 78.3 g (4.5 g fish per liter of water). The groupers were exposed to their respective treatment conditions for 11 weeks. The groupers were fed ad libitum 3-4 times per week with whole, freshly sacrificed spiny chromis (sacrificed using cerebral percussion). Each morning of the behavioral experiments (three consecutive days), the flow-through water in the predator tanks was turned off for 3 h (the air stone ensured dissolved oxygen remained >85% air saturation)

Table 1	Water chemistry data for the four sump systems (Control 1 and
2; High C	O ₂ 1 and 2) which each supplied three holding tanks during the
CO ₂ expo	osure period (May 7-July 27) prior to experiments. pCO2 was
measured	every 1-4 days, temperature was logged using iButton data-

loggers (one sample per 30 min), and alkalinity was measured on two occasions. pH_{TOTAL} was calculated using CO_2calc. Data are presented as mean \pm SD, and the seasonally dependent range in temperature is given in parentheses

Sump system	pCO_2 (µatm)	Temperature (°C)	Alkalinity (umolkg-1)	pH _{tot} (calc.)
Control 1	418 ± 13	24.5 ± 1.4 (21.4–30.2)	2353 ± 5	8.04 ± 0.00
Control 2	428 ± 13	24.6 ± 1.6 (21.4–30.1)	2363 ± 6	8.03 ± 0.01
Mean control	424 ± 13	24.6 ± 1.5	2358 ± 8	8.03 ± 0.01
High CO ₂ 1	1028 ± 101	24.4 ± 1.4 (21.3–30.3)	2353 ± 7	7.70 ± 0.03
High CO ₂ 2	1008 ± 144	$24.6 \pm 1.5 \ (21.4 - 29.9)$	2356 ± 4	7.71 ± 0.04
Mean high CO_2	1012 ± 137	24.5 ± 1.5	2355 ± 6	7.70 ± 0.04

before 200 mL of predator water was collected from each tank. This protocol was designed to achieve a predator cue concentration comparable to that used in previous studies (Table 2). At the same time, seawater samples without predator cues (sham water) from the control and high CO_2 treatments were collected from the sumps. All water samples were kept refrigerated throughout the day and used within 8 h. Subsamples of 5 mL, prepared in syringes, were warmed to ambient water temperature over ~90 min prior to use in experiments.

Experimental design

Following 80-82 days of exposure to present-day or elevated CO₂ (on July 25–27, 2015), we commenced experiments to investigate whether spiny chromis exposed to high CO₂ differed from control fish in their routine activity levels as well as in their behavioral response to predator chemical cues. We did this by measuring routine activity levels of individual fish before and after the injection of sham/predator water. Nine white opaque plastic arenas (diameter 11 cm; water volume 500 mL) containing water (23.3 \pm 0.27 °C, >98% air saturation) of the appropriate exposure treatment were placed in a 3×3 arrangement (Electronic Supplementary Material, Fig. S1), in a shallow reservoir aquarium receiving flow-through water to maintain temperature. The location of high CO₂ and control fish within the 3×3 arena grid was randomized for each trial. Each arena was fitted with an ~80 cm length of thin silicone tubing, one end of which was glued vertically to the inside of the arena such that it extended 1 cm below the surface of the water (Electronic Supplementary Material, Fig. S1). The other end of the tube was anchored ~ 20 cm outside the reservoir aquarium to enable the injection of cue water (seawater with or without predator cues). A curtain surrounding the reservoir aquarium minimized visual disturbances and prevented the fish from seeing the experimenter, including during the injection of sham/predator water.

The spiny chromis (TL mean \pm SD CO₂ 42.7 \pm 4.7 mm, control 41.6 \pm 4.0 mm; two-sample *t* test $t_{71} = 1.07$; P = 0.288) were placed individually into the experimental arenas and left undisturbed in moderate ambient light. After 90 min, 5 mL of sham water or 5 mL of predator cue water (Table 2) was slowly injected into each arena using 50-mL syringes connected to the end of the tubes extending outside the reservoir aquarium. All arenas were injected with cue within a 2-min period, and the fish were then left undisturbed for an additional 30 min. The fish in all nine arenas were monitored simultaneously throughout the experimental period using a FireWire camera (Dragonfly 2, Point Gray, Richmond, BC, Canada) mounted approximately 1 m above the arenas. To minimize observer bias, blinded methods were used when extracting and analyzing all behavioral data from the footage. We used tracking software (ViewPoint Zebralab; ViewPoint, Lyon, France) to automatically quantify swimming distance and duration (per minute) of each individual fish in real-time (as per Sundin and Jutfelt 2016) for 30 min prior to cue injection and 30 min after cue injection. Based on pilot runs, we iteratively set the swimming threshold to all movements exceeding 0.5 body lengths per second (BL s^{-1}). The remaining time reflects time spent inactive (i.e., "swimming duration" is the inverse of "inactivity duration"). The use of tracking software requires high contrast between the fish and the background, as documented in several previous studies that have used similar approaches to quantify behavioral responses, including in the context of elevated CO₂ (Bignami et al. 2013, 2014; Maneja et al. 2013; Hamilton et al. 2014; Maneja et al. 2015; Ou et al. 2015; Duteil et al. 2016; Sundin and Jutfelt 2016). The protocol was replicated eight times (trials), always using new fish, and randomizing exposure treatments and sham/predator water injections. The following sample sizes were obtained: N CO₂sham water injection = 16; CO₂-predator cue injection = 21; Control-sham water injection = 16; Control-predator cue injection = 19. After each trial, the fish were measured for TL and returned to labeled holding tanks. Dissolved oxygen of the water in the arenas declined to a minimum of $76.7 \pm 4.4\%$ of air saturation by the end of each trial.

Table 2the exp(Bioma(Biomawater (and caldescrib	Summary of meth erimental arena. Giv ss; see footnote), pre fime flow off), calcu culated apparent pre culated in the "Methods"	ven and details ven are the reft cdator tank watu dator tue cont cdator cue cont section. Cases	i of the preparation erences (Ref.), pre er volume (Vol.), c: concentration (Tot centration (Appare where insufficient	of predator chemic dator species (Spe alculated predator l al conc.), method c int conc.). Apparer t information was g	al cues used cies), numbe biomass per l of sample sto ut predator cu given are den	in previous studies r of individual pr iter of water (Rel. rage (Storage), vo ue concentration v oted NA	s, focusing on stuc edators per tank (biomass), duratio lume injected in t was calculated us	lies of coral reef No. pred.), pred n that the flow-t he experimental ing "Rel. bioma	fishes where t dator body len hrough was ce arena relative ass," "Time fl	he cue was inject gth (Length), pr ased prior to coll to arena volume ow off," and "fr	ed directly into edator biomass ecting predator (Inject: arena), ject: arena" as
Ref.	Species	No. pred.	Length (mm)	Biomass (g) [#]	Vol. (L)	Rel. biomass $(g L^{-1})$	Time flow off (h)	Total conc. $(g L^{-1})$	Storage	Inject: arena (mL)	Apparent conc. (g L^{-1})
**	C. urodeta	5	$143; 152 \& 129: 165^{\mathrm{T}}$	111.8 & 111.7	25	4.47 & 4.47	3	13.42 & 13.40	Fridge	5:500	0.13 & 0.13
1	P. fuscus	1	$57.8\pm5.6^{\rm S}$	2.3	9	0.39	96	37.58	Freezer	60:12,000	0.19
2	P. fuscus	2	57; 79 ^S	2.2; 6.2	10	0.85	9	5.07	Freezer	30:13,000	0.01
	S. dermatogenys	2	93; 102 ^S	16.0; 21.6	10	3.75	9	22.52	Freezer	30:13,000	0.05
	C. batuensis	2	124; 86 ^S	42.9; 14.0	10	5.69	9	34.13	Freezer	30:13,000	0.08
Э	C. cyanostigma	2	270; 250 & 290; 250 ^T	488.4 & 554.8	30	16.28 & 18.49	56	911.61 & 1035.57	NA^g	$20:14000^{f}$	1.44
4	P. fuscus	2	65; 71 ^S	3.4; 4.5	70	0.11	NA^{d}	I	<20 min	20:20,000	I
5	T. lunare	1	NA	I	30	I	12	I	NA	NA:15,000	I
	S. dermatogenys	1	NA	I	30	Ι	12	Ι	NA	NA:15,000	Ι
9	C. boenak	NA	$170.2\pm11.4^{\rm S}$	134.9	15	I	24	Ι	<20 min	60:15,000	I
7	P. fuscus	$\approx 6-8^{a}$	NA	I	32	I	NA	Ι	NA	60:NA	I
8	P. fuscus	$\approx 6-8^{a}$	NA	Ι	32	I	NA	Ι	NA	60:NA	Ι
9	C. argus	1	NA	I	70	I	2	I	NA^g	60:7800	I
10	P. fuscus	1	60 & 70 ^S	2.6 & 4.3	6	0.29 & 0.47	18:00—O/N	3.52 D. 5 60°	NA^g	20:9000	0.01 & 0.01
11	T. lunare	≥2 ^b	NA	I	Adj.	1.1 ^c	N/O	$\frac{\infty}{13.20^{\circ}}$	NA	30:13,000	0.03

0.03

30:13,000

30:13,000

 13.20^{e} 13.20^e 13.20^e

0.03

30:13,000

1.55 0.03

2:10

<1 day NA^{g} NA^{g} NA^{g}

7.74

0.65

6.5

L L 1

NA

 $\stackrel{|\geq 2^b}{\cong} \stackrel{|\geq 2^b}{\cong}$

T. hardwicke

T. lunare

13

C. batuensis

Ι

I

 1.1^{c} 1.1° 1.1°

NA

0.03 0.03 0.03

NA NA NA NA

ī ī I

> 30:20,000 30:20,000

NA

ī I I

ī T

NA NA NA NA NA NA NA NA NA

T. hardwicke

C. batuensis

S. dermatogenys

P. fuscus P. fuscus

T. lunare

12

P. fuscus

Adj. Adj. Adj. 20 20 20 Adj. Adj.

0/N 0/N 0/N 0/N 0/N 0/N

30:13,000 30:13,000 30:20,000

30:13,000

 13.20^{e} 13.20^{e} 13.20^{e}

 1.1^{c}

 $1.1^{\rm c}$ 1.1^{c}

I

L T

def. Species N P. fuscus 2 5 P. fuscus 8 7 Standard species names Cephalo 8 Maryside by authors to reach 1 7/N stated by authors as "overnig 9/N stated by authors as "overnig 8 Manassa et al. (2013a); 9 Manas: 8 Manassa et al. (2015) * Present study * * Dresent study * * Measured in the present study bu	lo. pred. 2 ^b JA <i>pholis urodet</i> <i>T</i> total length targeted "con tht" or as "18 ; 2 Mitchell et sa et al. (2013	Length (mm) NA 74.2 \pm 0.9 ⁸ 74.2 \pm 0.1 ¹ a. <i>Pseudochromis</i>) a. <i>Pseudochromis</i>) a. (2011); 3 Bosig b); 10 Mitchell and	Biomass (g) [#] - 5.1 <i>fuscus, Symodus derm</i> <i>fuscus, Symodus derm</i>	Vol. (L) Adj. 32 <i>aatogenys, Ci</i> trari et al. (20	Rel. biomass (g L ⁻¹) 1.1 [°] – <i>pris batuensis, Cepl</i> 12a); 5 Lönnstedt e et al. (2013); 12 Ch	Time flow off (h) O/N 16 <i>idopholis cyanosi</i> al. (2012); 6 Mar vers et al. (2014);	Total conc. (g L ⁻¹) 13.20 ^e - <i>igma, Thalassom</i> assa and McCorr <i>I3</i> Atherton and	Storage NA ^B NA <i>ia lunare, Cepli</i> mick (2012); 7 McCormick (2	Inject: arena (mL) 30:13,000 30:32,000 <i>alopholis boenak</i> , Manassa and McC (015);14 Mitchell ¢	Apparent conc. (g L ⁻¹) 0.03 - <i>Cephalopholis</i> tal. (2013) tal. (2013)
<i>P. fuscus</i> ≥ <i>P. fuscus</i> N Jenus and species names <i>Cephalo</i> <i>argus, Talassoma hardwicke</i> <i>inperscripted S</i> standard length, <i>1</i> <i>Adj</i> : adjusted by authors to reach 1 <i>ON</i> stated by authors to reach 1 <i>ON</i> stated by authors as "overnig <i>I</i> Holmes and McCormick (2010); <i>B</i> Manassa et al. (2013a); <i>9</i> Manas; Ramasamy et al. (2015) <i>P</i> Present study <i>P</i> Measured in the present study bu (25)	2 ^b IA <i>pholis urodet</i> <i>t</i> total length targeted "con targeted "con tht" or as "18 ; 2 Mitchell et sa et al. (2013	NA 74.2 \pm 0.9 ^S <i>a, Pseudochromis J</i> <i>a, Pseudochromis J</i> centration" of 1.1 { :00 and overnight" al. (2011); 3 Bosig b); <i>10</i> Mitchell and	- 5.1 <i>fuscus, Synodus dern</i> g L ⁻¹ water , ter et al. (2012); <i>4</i> Fei i McCormick (2013)	Adj. 32 <i>iatogenys, Cc</i> rrari et al. (20	1.1 [°] – <i>iris batuensis, Cepl</i> 12a); 5 Lönnstedt e et al. (2013); 12 Ch	O/N 16 <i>alopholis cyanosi</i> al. (2012); 6 Mar vers et al. (2014);	13.20 [°] – <i>tigma, Thalassom</i> assa and McCorr <i>13</i> Atherton and <i>1</i>	NA ^g NA <i>:a lunare, Ceph</i> mick (2012); 7 McConmick (2	30:13,000 30:32,000 <i>talopholis boenak</i> , Manassa and McC (015);14 Mitchell 6	0.03 – <i>Cephalopholis</i> ormick (2013); tal. (2015); <i>1</i> 3
Jenus and species names <i>Cephalo</i> <i>trgus, Talassoma hardwicke</i> <i>inperscripted S</i> standard length, <i>1</i> <i>Mj.</i> adjusted by authors to reach t <i>3/N</i> stated by authors as "overnig <i>1</i> Holmes and McCormick (2010); <i>3</i> Manassa et al. (2013a); <i>9</i> Manas; Ramasamy et al. (2015) <i>i</i> Present study <i>i</i> Measured in the present study bu (.25	<i>pholis urodet</i> <i>T</i> total length targeted "con tht" or as "18 ; 2 Mitchell et sa et al. (2013	<i>a, Pseudochromis J</i> centration" of 1.1 { :00 and overnight" al. (2011); 3 Bosig b); <i>10</i> Mitchell and	<i>fuscus, Synodus derm</i> g L ⁻¹ water , er et al. (2012); <i>4</i> Fei i McCormick (2013):	<i>iatogenys, Cc</i> rrari et al. (20	<i>ris batuensis, Cepl</i> 12a); <i>5</i> Lönnstedt e et al. (2013); <i>12</i> Chi	alopholis cyanost al. (2012); 6 Mar vers et al. (2014);	t <i>igma, Thalassom</i> assa and McCorr <i>13</i> Atherton and l	a lunare, Ceph nick (2012); 7 McConnick (2	<i>talopholis boenak</i> , Manassa and McC (015); 14 Mitchell 6	<i>Cephalopholis</i> ormick (2013); t al. (2015); <i>15</i>
<i>uperscripted S</i> standard length, <i>T</i> <i>Idj</i> : adjusted by authors to reach t <i>DN</i> stated by authors as "overnig <i>N</i> holmes and McCormick (2010); 8 Manassa et al. (2013a); 9 Manas; 8 mansamy et al. (2015) i Present study ' Measured in the present study bu (.25	7 total length targeted "cond pht" or as "18 5 2 Mitchell et sa et al. (2013	centration" of 1.1 { :00 and overnight" al. (2011); 3 Bosig b); 10 Mitchell and	g L ⁻¹ water , ; ger et al. (2012); <i>4</i> Fer i McCormick (2013);	rrari et al. (20 ; <i>11</i> Mitchell 6	12a); 5 Lönnstedt e et al. (2013); 12 Chi	. al. (2012); 6 Mar vers et al. (2014);	assa and McCorr 13 Atherton and 1	nick (2012); 7 McCormick (2	Manassa and McC 015); 14 Mitchell 6	armick (2013); t al. (2015); <i>15</i>
<i>ldj</i> . adjusted by authors to reach t <i>2/N</i> stated by authors as "overnig ' Holmes and McCormick (2010); 8 Manassa et al. (2013a); <i>9</i> Manas: Ramasamy et al. (2015) Present study ' Present study ' Measured in the present study bu 1.25	htt" or as "18 htt" or as "18 ; 2 Mitchell et sa et al. (2013	sentration" of 1.1 { :00 and overnight" al. (2011); 3 Bosig b); 10 Mitchell and	g L ⁻¹ water , ; ger et al. (2012); 4 Fei il McCormick (2013);	rrari et al. (20 ; 11 Mitchell 6	12a); 5 Lönnstedt e et al. (2013); <i>12</i> Chi	.al. (2012); 6 Mar vers et al. (2014);	lassa and McCorr <i>13</i> Atherton and]	nick (2012); 7 McCormick (2	Manassa and McC 015); <i>14</i> Mitchell e	ormick (2013); t al. (2015); <i>15</i>
<i>2/N</i> stated by authors as "overnig" ' Holmes and McCormick (2010); <i>3</i> Manassa et al. (2013a); <i>9</i> Manas; <i>8</i> Manasamy et al. (2015) 'Present study ' Measured in the present study bu (.25	ht" or as "18 ; 2 Mitchell et sa et al. (2013	:00 and overnight" al. (2011); 3 Bosig b); 10 Mitchell and	, ;er et al. (2012); 4 Fet 1 McCormick (2013);	rari et al. (20 ; 11 Mitchell (12a); 5 Lönnstedt ei et al. (2013); <i>12</i> Chí	. al. (2012); 6 Man vers et al. (2014);	assa and McCorn 13 Atherton and 1	nick (2012); 7 McCormick (2	Manassa and McC 015); <i>14</i> Mitchell e	ormick (2013) t al. (2015); <i>11</i>
¹ Holmes and McCormick (2010); 8 Manassa et al. (2013a); 9 Manass Ramasamy et al. (2015) ¹ Present study ⁴ Measured in the present study bu 1.25	, 2 Mitchell et sa et al. (2013	al. (2011); <i>3</i> Bosig b); <i>10</i> Mitchell and	er et al. (2012); 4 Fer I McCormick (2013);	rari et al. (20 ; 11 Mitchell e	12a); 5 Lönnstedt ef et al. (2013); <i>12</i> Chi	al. (2012); 6 Man vers et al. (2014);	iassa and McCorn 13 Atherton and l	nick (2012); 7] McCormick (2	Manassa and McC (015); 14 Mitchell e	ormick (2013) t al. (2015); <i>1</i> :
Present study ¹ Measured in the present study bu										
Measured in the present study bu										
	ıt calculated fi	or all other studies	based on length-mas	s relationship	s in Froese et al. (2))14) and adjusting	standard length t	to total length,	where necessary, u	sing a factor of
'Stated in the papers as "approxir.	nately 6–8"									
Stated in the papers as "at least t	"wo"									
Predator mass per liter specified	in the paper									
¹ Authors stated 60% of the water	volume was	changed per day								
Assuming that the author statem	ents "18:00 a	nd overnight" and	"overnight" are equi	ivalent to 12	h					
Stated that 10 mL from each of t	wo predator t	anks was injected	into a 14-L experime	ental arena, w	rhich gives one app	arent predator cue	e concentration			
¹ Details not provided in original 1	reference but	stated as "consister	ntly available," "coll	ected just pri	or to experiment,"	"collected prior tc	start," or "used	the following i	morning"	

Statistical analysis

We tested whether there was an effect of CO₂ treatment on routine activity (i.e., activity levels before cue injection) using analysis of variance (ANOVA). Mean swimming duration and distance (i.e., all movements exceeding 0.5 BL s⁻¹) for the 30min period following acclimation were used as the response variables (i.e., duration and distance, respectively), while treatment (CO₂ or control) was used as a fixed effect. Fish length was initially included as a covariate, but it was found to have no effect on swimming activity (Length_[duration] F_{1} , $_{64} = 2.59$, P = 0.113; Length_[distance]: F_{1} , $_{64} = 0.29$, P = 0.595) and was therefore removed from the models and is not discussed further.

The predator cue concentration injected into the experimental arenas was estimated for the present and previous studies (where sufficient information was provided, and/or by inferring data) in order to enable comparisons across studies (Table 2). Calculations were conducted as follows (with values from the present study given in parentheses as an example). Predator biomass in the predator holding tanks was established (112 g/ $25 \text{ L} = 4.5 \text{ g L}^{-1}$) and multiplied by the duration for which the flow-through water was ceased prior to obtaining a water sample (4.5 g $L^{-1} \times 3 h = 13.4 g L^{-1}$). This total concentration was multiplied by the amount of cue injected into the experimental arena (13.4 g $L^{-1} \times 0.005 L = 0.067 g$) and then divided by the volume of the experimental arena (0.067 g/ $0.5 \text{ L} = 0.13 \text{ g L}^{-1}$) to provide an estimate that is herein referred to as the "apparent predator cue concentration." In instances where only predator length was provided in previous studies, we estimated mass based on length-mass relationships in Froese et al. (2014), after adjusting standard length to total length using a factor of 1.25 where necessary. While these estimates contain several assumptions (e.g., that predator cue increases linearly over time upon cessation of flow-through water) and do not represent true concentrations, they provide useful information for comparing across studies and highlighting some of the critical details that we urge authors to include in future papers in this field (Table 2).

To investigate the response to sham/predator cues, we calculated the difference in mean swimming duration and distance between the pre- and post-injection periods (i.e., mean duration and distance during the 30-min pre-injection period subtracted from mean duration and distance during the 30-min post-injection period; 2 min was excluded between these periods to allow for cue injections). This analytical approach accommodates for inter-individual variation in fish activity levels and allows for easier comparisons with previous studies that have used a similar approach (e.g., Ferrari et al. 2011a, b, 2012a, b; Lönnstedt et al. 2013; Chivers et al. 2014). Differences in mean swimming duration and distance were the response

variables in ANOVAs, whereas treatment (CO₂ or control), cue type (sham water or predator water), and their interaction were included as fixed effects. Since the predator cue was prepared in the morning of each day and refrigerated for use throughout the day, we further included time of day as a factor, and the interaction between time of day and cue type, to investigate possible degradation of the predator cue. Fish length was again included as a covariate and had no effect on swimming activity (Length_[duration] F_{1} , $_{57} = 0.80, P = 0.374$; Length_[distance] $F_{1, 57} = 1.45,$ P = 0.234) and was therefore removed from the models. For the minimal-adequate model, non-significant interactions (α level 0.10) were dropped using stepwise backward exclusion. Dropped interactions were again included in the final model, one at a time, to verify that they did not have a significant effect. Since none of the interactions were found to be significant (P > 0.30), the final model contained cue, treatment, and time of day (final model ANOVA, duration $F_{6, 56} = 2.81$, P = 0.019, adjusted $R^2 = 0.15$, $\Delta AIC_c = 8$ compared with full model; distance $F_{6,56} = 3.17, P = 0.010, \text{ adjusted } R^2 = 0.17, \Delta AIC_c = 7$ compared with full model). In all cases, the data met the assumptions of residual normal distribution and homogeneity of variance (Shapiro-Wilk W > 0.9, P > 0.50, Levene $F_1 > 0.2$, P > 0.20, respectively). JMP 11 (SAS Institute Inc., Cary, NC, USA) was used for the statistical analyses.

Results

In contrast with our prediction, there were no differences in routine activity levels between CO₂ treatment groups during the 30-min period prior to injection of the cue (Treatment_[duration] $F_{1, 65} = 1.27$, P = 0.264; Treatment_[distance] $F_{1, 65} = 0.30$, P = 0.588; Fig. 1). Combining both CO₂ treatment groups and both cue type groups (sham and predator cue), fish spent 20.4 ± 0.9 s min⁻¹ swimming at >0.5 BL s⁻¹ (mean ± S.E.), and they swam a distance of 547 ± 28 mm min⁻¹ during the 30 min prior to cue injection (Fig. 1).

While comparisons of predator cue "concentrations" across studies are difficult due to a general lack of details (Table 2), through our calculations of apparent predator cue concentrations, we established that the levels we used were at least as high as those estimated for the majority of previous studies (0.13 g L⁻¹; Table 2). As predicted, cue type had an effect on the swimming duration and distance of spiny chromis, whereby activity decreased in fish receiving the predator cue injection but did not differ from pre-injection levels in fish receiving the sham water injection (Cue type_[duration] $F_{1, 64} = 7.88$, P = 0.007; Cue type_[distance] $F_{1, 64} = 6.42$, P = 0.014; Fig. 1). This response was not dependent on CO₂ treatment, since the cue-induced changes in swimming duration and distance were the same in





Fig. 1 Mean \pm SE routine swimming duration (a) and swimming distance (b) of juvenile spiny chromis (*Acanthochromis polyacanthus*) after being reared for 3 months in control (~420 µatm, *black bars*) or high CO₂ (~1000 µatm, *gray bars*) treatments at 23.3 \pm 0.27 °C. Values were calculated during the 30-min period prior to cue injection

the control and high CO₂ groups (Treatment_[duration] F_{1} , ₆₄ = 0.14, P = 0.709; Treatment_[distance] F_{1} , ₆₄ = 0.48, P = 0.491; Fig. 2). Pooling data from both CO₂ treatment groups, mean swimming duration after the predator cue injection was 85% of the duration before the injection (mean ± SE duration before 21.0 ± 1.2 s min⁻¹, duration after 17.8 ± 1.1 s min⁻¹) and swimming distance after predator cue injection was 83% of the distance before the injection (mean ± SE distance before 553 ± 39 mm min⁻¹, distance after 460.4 ± 32.0 mm min⁻¹).

Time of day had a significant effect on the mean difference in activity pre- vs. post-injection (Time of day_[duration] $F_{5,}$ $_{64} = 4.57$, P = 0.001; Time of day_[distance] $F_{5, 64} = 5.11$, P = 0.001). However, the lack of an interaction between cue type and time of day (Time of day * Cue type_[duration] $F_{5,}$ $_{59} = 0.71$, P = 0.621; Time of day * Cue type_[distance] $F_{5,}$ $_{59} = 1.28$, P = 0.283) suggests that the time of day effect was unrelated to sequential changes in the predator chemical cue during refrigeration and re-warming. This lack of interaction also suggests that fish were not responding differently as the day progressed (e.g., due to potentially increased hunger levels). Indeed, Tukey's HSD post-hoc tests revealed that one

Fig. 2 Mean \pm SE change in swimming duration (a) and swimming distance (b) of juvenile spiny chromis (*Acanthochromis polyacanthus*) between the 30-min pre- and post-injection periods, where injections were sham (non-predator) water, or water-containing predator cues. Fish had been reared for 3 months in control (~420 µatm, *black bars*) or high CO₂ (~1000 µatm, *gray bars*) treatments at 23.3 \pm 0.27 °C (inserted illustration: flagtail grouper, *Cephalopholis urodeta*; George Henry Ford, https:// commons.wikimedia.org/wiki/File%3ACephalopholis urodeta.jpg)

specific trial (trial 7 on day 2 (July 26)), rather than a specific time of day, elicited the significant effect of time of day. Excluding that trial confirmed our initial results of an effect of cue type on the difference between pre- vs. post-injection activity levels (Cue type_[duration] $F_{1, 60} = 8.67$, P = 0.005; Cue type_[distance] $F_{1, 60} = 8.47, P = 0.005$). All other factors, including time of day and interactions, remained non-significant. Time of day had a similar effect on routine activity levels (pre-stimulus activity) (Time of day_[duration] $F_{5, 65} = 2.41$, P = 0.046; Time of day_[distance] $F_{5, 65} = 2.86$, P = 0.021), but again this was caused by lower activity in the same trial as indicated above. Correspondingly, CO2 treatment remained non-significant in the analysis of pre-stimulus routine activity levels when excluding the trial with anomalously low activity (Treatment_[duration] $F_{1, 61} = 2.15$, P = 0.148; Treatment_[distance] $F_{1, 61} = 0.70, P = 0.405).$

Discussion

Several aspects of predator avoidance behavior have been reported to be negatively affected by increased levels of CO₂. These include loss of behaviors typically triggered by visual and auditory predator stimuli (Simpson et al. 2011; Ferrari et al. 2012b; Lönnstedt et al. 2013), and attraction to, rather than avoidance of, predator chemical cues (Dixson et al. 2010; Munday et al. 2010, 2012, 2013; Nilsson et al. 2012). Impaired predator avoidance should result in reduced survival in fishes exposed to high CO2, which has indeed been reported in experiments in which coral reef fish were exposed to elevated CO₂ for 4-12 days and then tested in control water (Munday et al. 2010; Ferrari et al. 2011a, b; Munday et al. 2012; Chivers et al. 2014). In addition, increased levels of CO₂ have been reported to increase activity in several species of coral reef fishes (Munday et al. 2010, 2013; Cripps et al. 2011; Devine et al. 2012) with some exceptions (Ferrari et al. 2012a; Nowicki et al. 2012). Non-coral reef fishes appear to be more robust, as several papers have reported no effect of elevated CO₂ on activity in a range of species (e.g., Bignami et al. 2013, 2014; Maneja et al. 2013, 2015; Hamilton et al. 2014; Sundin and Jutfelt 2016).

The present study contrasts with the majority of reports on coral reef fishes by showing no effect of elevated CO₂ on routine activity levels or on the behavioral response to predator cues. Specifically, fish exposed to high CO₂ for 80-82 days had a similar routine activity level as control fish reared at present-day CO₂ levels, and they decreased their activity level to the same extent as the control fish when exposed to predator chemical cues. The decreased duration spent swimming after the predator cue injection indicates that the fish spent more time moving slowly (<0.5 BL s⁻¹) and/or being immobile, which is a typical response to predator cues (Kelley 2008 and references therein). While a 15% reduction in swimming duration and 17% reduction in swimming distance may seem relatively small, it has been reported that damselfish larvae, Pomacentrus wardi, exposed to 850 µatm for 4 days and subsequently transplanted to patch reefs in situ suffer nine-fold higher predation mortality as a consequence of having 24% higher activity levels (~0.6 cm min⁻¹ higher) and moving an average of 0.8 cm further from shelter than control larvae (Munday et al. 2010).

The discrepancy between our findings and the majority of previous reports on tropical damselfishes, including *A. polyacanthus*, could be due to several factors, including differences in experimental design, exposure times, the high precision and low vulnerability to biases in the present study due to the use of video-tracking software (Egan et al. 2009; Holman et al. 2015), differences in behavioral responses between and within species (Clements and Hunt 2015), and/or publication and reporting biases (i.e., the observation that

studies reporting negative results are typically more difficult to publish) (Browman 1999; Dwan et al. 2013). We expand on some of these points below.

To the best of our knowledge, the present study is the first to use automated tracking software to quantify the activity responses of a coral reef fish to predator chemical cues in the context of ocean acidification (although several studies have examined different contexts and species; e.g., Bignami et al. 2013, 2014; Hamilton et al. 2014; Sundin and Jutfelt 2016). Previous studies on coral reef fishes have typically assessed activity by manually scoring, in real-time with the naked eye, the number of lines crossed by a fish in an aquarium (Ferrari et al. 2011a, 2012a; Nowicki et al. 2012; Lönnstedt et al. 2013; Munday et al. 2014). While not diminishing the value of those studies, video tracking provides more precise data of higher validity and objectivity (Egan et al. 2009). The use of automated video tracking ensures that the scoring of activity is not subject to observer biases (e.g., confirmation bias, which is the unintentional preference to detect and focus on outcomes that confirm prior beliefs; Nickerson 1998; Holman et al. 2015), and it provides a database of visual evidence.

There are additional methodological differences between our study and previous studies examining the effect of CO₂ on fish behavior. For example, several previous studies have investigated activity levels and responses to chemical cues in combination with a feeding event (Ferrari et al. 2011a, 2012b; Chivers et al. 2014). Although fish were not fed during the behavioral observations in our experiment, an effect of the predator cue was detected, so we do not view this as a confounding factor. Indeed, CO2 treatment effects on swimming activity have been reported in previous studies in which food was withheld during behavioral observations (e.g., Munday et al. 2013, 2014; Pimentel et al. 2014; Rossi et al. 2015). We also used a relatively small behavioral arena in relation to fish size compared with some previous studies (e.g., Ferrari et al. 2011a, b), which may have restricted the magnitude of the change in activity of the fish in response to the predator cue. Whether a larger arena would have allowed for a greater reduction in movement following cue injection is unclear. Nevertheless, despite the smaller arenas, the reaction by the fish in our experiment was the same regardless of CO₂ acclimation treatment. Furthermore, in this study, we used captivebred fish (from a public aquarium), similar to some previous studies (e.g., Rossi et al. 2015; Pistevos et al. 2017) but in contrast with other studies that have used wild fish or fish 1-2 generations removed from the wild (e.g., Munday et al. 2010). Based on existing literature, the reported effects of elevated CO₂ do not seem to differ between wild and captive-bred fish, yet clarity on this point must await further research.

It could be argued that the absence of a CO_2 treatment effect in the present study is due to *A. polyacanthus* being

more resilient to high CO₂ compared with other coral reef damselfishes for which dramatic behavioral impairments have been reported (reviewed in Clements and Hunt 2015). Indeed, Munday et al. (2011) reported no effects of CO_2 on juvenile growth, survival, skeletal development, or otolith morphometrics after rearing newly hatched A. polyacanthus for 21 days in pH_{NBS} 7.83. There also appears to be no detrimental effect on the oxygen uptake rates of this species following 17 days in pH_{NBS} 7.87 (Rummer et al. 2013). However, evidence against the idea of A. polyacanthus being resilient to high CO₂ stems from other studies of this species that have reported profound effects. For example, Chung et al. (2014) reported an altered retinal response in this species after 6-7 days of exposure to pH_{NBS} 7.87. Furthermore, Welch et al. (2014) reported that 40-45 days of exposure to pH_{NBS} 7.85 resulted in reduced behavioral lateralization and attraction to (rather than avoidance of) chemical alarm cues extracted from the skin of conspecifics. Disparate findings within a species have also been reported for the tropical damselfish P. wardi. For example, Bender et al. (2015) reported that fish did not show any visible changes in behavior and did not change their food selectivity when exposed to pH_{TOTAL} 7.85 in combination with warming (+4.1 °C) for 4 weeks. Other studies on the same species following 4-7 days of exposure to similar CO₂ levels reported that CO₂ exposure induces riskier behavior, causes attraction to predator cues (93% of time spent in predator cues), increases activity, and shifts behavioral lateralization from a right to a left bias (Munday et al. 2010 pCO₂ 850 µatm; Munday et al. 2012 pH_{NBS} 7.98; and Domenici et al. 2014 pH_{NBS} 7.87). Hence, it seems that elevated CO₂ may or may not cause impairments within a species depending on which behavioral and physiological traits are investigated, but this is a difficult conclusion to make in the absence of standardized replication studies.

In sum, we found that long-term exposure to elevated CO_2 did not alter routine activity levels or the behavioral response to predator chemical cues in *A. polyacanthus*. This study thus adds to the growing literature reporting no behavioral effects in fishes following acclimation to elevated CO_2 (Browman 2016). Contrasting findings among studies using the same species demonstrate the importance of replication as a prerequisite to establishing a consensus on how fish behavior may be affected by ocean acidification. Achieving such replication will rely on transparency and the use of robust and standardized methods.

Data availability

The dataset supporting this manuscript is available as electronic supplementary material and publicly archived in the repository figshare, following best practices (Roche et al. 2015). https://figshare.com/s/64d7082241deda6d4bd4

Acknowledgments This work was funded by the Swedish Research Council Formas (JS 2013-947; FJ 2009-596), the Swedish Research Council VR (MA 637-2014-449; FJ 621-2012-4679), the Wenner-Gren Foundation (JS, MA), the Society for Experimental Biology and Company of Biologists Travel Grants (JS JEBTF-150422), the Fulbright Foundation (MA), and through an Australian Endeavour Research Fellowship (GDR). Further support and facilities were provided by the Australian Institute of Marine Science (TDC). Special thanks to the SeaSim staff at the Australian Institute of Marine Science for logistical support. Two anonymous reviewers provided comments to help clarify some points in the manuscript.

Author contributions All authors designed the experiments. JS, MA, FMG, GDR, and TDC performed the experiments. FJ contributed vital equipment. JS and MA analyzed the data. JS wrote the manuscript draft with significant contributions in addition to approval from all other authors.

Compliance with ethical standards

Ethical approval The experiments were performed according to current national legislation on Animal Welfare under license A1924 from the James Cook University Animal Ethics Committee.

Open Access This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made.

References

- Atherton JA, McCormick MI (2015) Active in the sac: damselfish embryos use innate recognition of odours to learn predation risk before hatching. Anim Behav 103:1–6
- Bender D, Champ CM, Kline D, Diaz-Pulido G, Dove S (2015) Effects of "reduced" and "business-as-usual" CO₂ emission scenarios on the algal territories of the damselfish *Pomacentrus wardi* (Pomacentridae). PLoS One 10:e0131442
- Bignami S, Sponaugle S, Cowen RK (2013) Response to ocean acidification in larvae of a large tropical marine fish, *Rachycentron canadum*. Glob Change Biol 19:996–1006
- Bignami S, Sponaugle S, Cowen RK (2014) Effects of ocean acidification on the larvae of a high-value pelagic fisheries species, mahi-mahi *Coryphaena hippurus*. Aquat Biol 21:249–260
- Bockmon EE, Dickson AG (2015) An inter-laboratory comparison assessing the quality of seawater carbon dioxide measurements. Mar Chem 171:36–43
- Bosiger YJ, Lönnstedt OM, McCormick MI, Ferrari MCO (2012) Learning temporal patterns of risk in a predator-diverse environment. PLoS One 7:e34535
- Browman HI (1999) Negative results. Mar Ecol-Prog Ser 191:301–309
- Browman HI (2016) Applying organized scepticism to ocean acidification research. ICES J Mar Sci 73:529–536
- Caldeira K, Wickett ME (2005) Ocean model predictions of chemistry changes from carbon dioxide emissions to the atmosphere and ocean. J Geophys Res-Oceans 110:1–12
- Chivers DP, McCormick MI, Nilsson GE, Munday PL, Watson S-A, Meekan MG, Mitchell MD, Corkill KC, Ferrari MCO (2014) Impaired learning of predators and lower prey survival under elevated CO₂: a consequence of neurotransmitter interference. Glob Change Biol 20:515–522

- Chung W-S, Marshall NJ, Watson S-A, Munday PL, Nilsson GE (2014) Ocean acidification slows retinal function in a damselfish through interference with GABA(A) receptors. J Exp Biol 217:323–326
- Clements JC, Hunt HL (2015) Marine animal behaviour in a high CO₂ ocean. Mar Ecol-Prog Ser 536:259–279
- Cornwall CE, Hurd CL (2016) Experimental design in ocean acidification research: problems and solutions. ICES J Mar Sci 73:572–581
- Crawford D, Harrison P (1997) Direct measurement of pCO_2 in cultures of marine phytoplankton: how good is the estimate from pH_{NBS} and single point titration of alkalinity? Mar Ecol-Prog Ser 158:61–74
- Cripps IL, Munday PL, McCormick MI (2011) Ocean acidification affects prey detection by a predatory reef fish. PLoS One 6:e22736
- Devine BM, Munday PL, Jones GP (2012) Homing ability of adult cardinalfish is affected by elevated carbon dioxide. Oecologia 168: 269–276
- Dickson AG (1984) pH scales and proton-transfer reactions in saline media such as sea water. Geochim Cosmochim Ac 48:2299–2308
- Dickson AG (1990) Standard potential of the reaction—AGCL(s)+1/ $2H_2(g) = AG(s)+HCL(aq)$ and the standard acidity constant of the ion HSO₄⁻⁻ in synthetic sea-water from 273.15-K to 318.15-K. J Chem Thermodyn 22:113–127
- Dixson DL, Munday PL, Jones GP (2010) Ocean acidification disrupts the innate ability of fish to detect predator olfactory cues. Ecol Lett 13:68–75
- Dixson DL, Jennings AR, Atema J, Munday PL (2015) Odor tracking in sharks is reduced under future ocean acidification conditions. Glob Change Biol 21:1454–1462
- Domenici P, Allan BJM, Watson S-A, McCormick MI, Munday PL (2014) Shifting from right to left: the combined effect of elevated CO₂ and temperature on behavioural lateralization in a coral reef fish. PLoS One 9:e87969
- Doney SC, Fabry VJ, Feely RA, Kleypas JA (2009) Ocean acidification: the other CO₂ problem. Annu Rev Mar Sci 1:169–192
- Dupont S, Dorey N, Thorndyke M (2010) What meta-analysis can tell us about vulnerability of marine biodiversity to ocean acidification? Estuar Coast Shelf S 89:182–185
- Duteil M, Pope E, Perez-Escudero A, de Polavieja G, Fürtbauer I, Brown M, King A (2016) European sea bass show behavioural resilience to near-future ocean acidification. R Soc Open Sci 3:160656
- Dwan K, Gamble C, Williamson PR, Kirkham JJ (2013) Systematic review of the empirical evidence of study publication bias and outcome reporting bias—an updated review. PLoS One 8:e66844
- Egan RJ, Bergner CL, Hart PC et al (2009) Understanding behavioral and physiological phenotypes of stress and anxiety in zebrafish. Behav Brain Res 205:38–44
- Ferrari MCO, Dixson DL, Munday PL, McCormick MI, Meekan MG, Sih A, Chivers DP (2011a) Intrageneric variation in antipredator responses of coral reef fishes affected by ocean acidification: implications for climate change projections on marine communities. Glob Change Biol 17:2980–2986
- Ferrari MCO, McCormick MI, Munday PL, Meekan MG, Dixson DL, Lönnstedt O, Chivers DP (2011b) Putting prey and predator into the CO₂ equation—qualitative and quantitative effects of ocean acidification on predator-prey interactions. Ecol Lett 14:1143–1148
- Ferrari MCO, Manassa RP, Dixson DL, Munday PL, McCormick MI, Meekan MG, Sih A, Chivers DP (2012a) Effects of ocean acidification on learning in coral reef fishes. PLoS One 7:e31478
- Ferrari MCO, McCormick MI, Munday PL, Meekan MG, Dixson DL, Lönnstedt O, Chivers DP (2012b) Effects of ocean acidification on visual risk assessment in coral reef fishes. Funct Ecol 26:553–558
- Froese R, Thorson JT, Reyes R (2014) A Bayesian approach for estimating length-weight relationships in fishes. J Appl Ichthyol 30:78–85
- Green L, Jutfelt F (2014) Elevated carbon dioxide alters the plasma composition and behaviour of a shark. Biol Lett 10:2014053

- Hamilton TJ, Holcombe A, Tresguerres M (2014) CO₂-induced ocean acidification increases anxiety in rockfish via alteration of GABA(A) receptor functioning. Proc R Soc B 281:20132509
- Heuer RM, Grosell M (2014) Physiological impacts of elevated carbon dioxide and ocean acidification on fish. Am J Physiol-Reg I 307: R1061–R1084
- Holbrook SJ, Schmitt RJ (2003) Spatial and temporal variation in mortality of newly settled damselfish: patterns, causes and co-variation with settlement. Oecologia 135:532–541
- Holman L, Head ML, Lanfear R, Jennions MD (2015) Evidence of experimental bias in the life sciences: why we need blind data recording. PLoS Biol 13:e1002190
- Holmes TH, McCormick MI (2010) Smell, learn and live: the role of chemical alarm cues in predator learning during early life history in a marine fish. Behav Process 83:299–305
- Ishimatsu A, Hayashi M, Kikkawa T (2008) Fishes in high-CO₂ acidified oceans. Mar Ecol-Prog Ser 373:295–302
- Kelley JL (2008) Assessment of predation risk by prey fishes. In: Magnhagen M, Braithwaite VA, Forsgren E, Kapoor BG (eds) Fish behaviour. Science Publishers, Enfield, pp 269–301
- Leduc AOHC, Munday PL, Brown GE, Ferrari MCO (2013) Effects of acidification on olfactory-mediated behaviour in freshwater and marine ecosystems: a synthesis. Philos T Roy Soc B 368:20120447
- Lönnstedt OM, McCormick MI, Meekan MG, Ferrari MCO, Chivers DP (2012) Learn and live: predator experience and feeding history determines prey behaviour and survival. Proc R Soc Lond B 279: 2091–2098
- Lönnstedt OM, Munday PL, McCormick MI, Ferrari MCO, Chivers DP (2013) Ocean acidification and responses to predators: can sensory redundancy reduce the apparent impacts of elevated CO₂ on fish? Ecol Evol 3:3565–3575
- Lueker TJ, Dickson AG, Keeling CD (2000) Ocean *p*CO₂ calculated from dissolved inorganic carbon, alkalinity, and equations for K1 and K2: validation based on laboratory measurements of CO₂ in gas and seawater at equilibrium. Mar Chem 70:105–119
- Manassa RP, McCormick MI (2012) Social learning and acquired recognition of a predator by a marine fish. Anim Cogn 15:559–565
- Manassa R, McCormick M (2013) Social learning improves survivorship at a life-history transition. Oecologia 171:845–852
- Manassa RP, McCormick MI, Chivers DP (2013a) Socially acquired predator recognition in complex ecosystems. Behav Ecol Sociobiol 67:1033–1040
- Manassa RP, McCormick MI, Chivers DP, Ferrari MCO (2013b) Social learning of predators in the dark: understanding the role of visual, chemical and mechanical information. Proc R Soc B 28:1765
- Maneja RH, Frommel AY, Browman HI, Clemmesen C, Geffen AJ, Folkvord A, Piatkowski U, Durif CMF, Bjelland R, Skiftesvik AB (2013) The swimming kinematics of larval Atlantic cod, *Gadus morhua* L., are resilient to elevated seawater pCO₂. Mar Biol 160: 1963–1972
- Maneja RH, Frommel AY, Browman HI, Geffen AJ, Folkvord A, Piatkowski U, Durif CMF, Bjelland R, Skiftesvik AB, Clemmesen C (2015) The swimming kinematics and foraging behavior of larval Atlantic herring (*Clupea harengus* L.) are unaffected by elevated pCO₂. J Exp Mar Biol Ecol 466:42–48
- Mitchell MD, McCormick MI (2013) Ontogenetic differences in chemical alarm cue production determine antipredator responses and learned predator recognition. Behav Ecol Sociobiol 67:1123–1129
- Mitchell MD, McCormick MI, Ferrari MC, Chivers DP (2011) Coral reef fish rapidly learn to identify multiple unknown predators upon recruitment to the reef. PLoS One 6:e15764
- Mitchell MD, McCormick MI, Chivers DP, Ferrari MCO (2013) Generalization of learned predator recognition in coral reef ecosystems: how cautious are damselfish? Funct Ecol 27:299–304
- Mitchell MD, Chivers DP, McCormick MI, Ferrari MC (2015) Learning to distinguish between predators and non-predators: understanding

Behav Ecol Sociobiol (2017) 71:108

the critical role of diet cues and predator odours in generalisation. Sci Rep 5:13918

- Moran D (2014) The importance of accurate CO₂ dosing and measurement in ocean acidification studies. J Exp Biol 217:1827–1828
- Munday PL, Dixson DL, McCormick MI, Meekan M, Ferrari MCO, Chivers DP (2010) Replenishment of fish populations is threatened by ocean acidification. P Natl Acad Sci USA 107:12930–12934
- Munday PL, Gagliano M, Donelson JM, Dixson DL, Thorrold SR (2011) Ocean acidification does not affect the early life history development of a tropical marine fish. Mar Ecol-Prog Ser 423: 211–221
- Munday PL, McCormick MI, Meekan M, Dixson DL, Watson SA, Chivers DP, Ferrari MCO (2012) Selective mortality associated with variation in CO₂ tolerance in a marine fish. Ocean Acidification 1:1–5
- Munday PL, Pratchett MS, Dixson DL, Donelson JM, Endo GGK, Reynolds AD, Knuckey R (2013) Elevated CO₂ affects the behavior of an ecologically and economically important coral reef fish. Mar Biol 160:2137–2144
- Munday PL, Cheal AJ, Dixson DL, Rummer JL, Fabricius KE (2014) Behavioural impairment in reef fishes caused by ocean acidification at CO₂ seeps. Nature Clim Change 4:487–492
- Nickerson RS (1998) Confirmation bias: a ubiquitous phenomenon in many guises. Rev Gen Psychol 2:175
- Nilsson GE, Dixson DL, Domenici P, McCormick MI, Sorensen C, Watson SA, Munday PL (2012) Near-future carbon dioxide levels alter fish behaviour by interfering with neurotransmitter function. Nature Clim Change 2:201–204
- Nowicki JP, Miller GM, Munday PL (2012) Interactive effects of elevated temperature and CO₂ on foraging behavior of juvenile coral reef fish. J Exp Mar Biol Ecol 412:46–51
- Ou M, Hamilton TJ, Eom J, Lyall EM, Gallup J, Jiang A, Lee J, Close DA, Yun S-S, Brauner CJ (2015) Responses of pink salmon to CO₂induced aquatic acidification. Nature Clim Change 5:950–955
- Pimentel M, Pegado M, Repolho T, Rosa R (2014) Impact of ocean acidification in the metabolism and swimming behavior of the

dolphinfish (Coryphaena hippurus) early larvae. Mar Biol 161: 725-729

- Pistevos JC, Nagelkerken I, Rossi T, Connell SD (2017) Ocean acidification alters temperature and salinity preferences in larval fish. Oecologia 183:545–553
- Ramasamy RA, Allan BJM, McCormick MI (2015) Plasticity of escape responses: prior predator experience enhances escape performance in a coral reef fish. PLoS One 10:e0132790
- Riebesell U, Fabry VJ, Hansson L, Gattuso J-P (2010) Guide to best practices for ocean acidification research and data reporting. Publications Office of the European Union, Luxembourg
- Roche DG, Kruuk LE, Lanfear R, Binning SA (2015) Public data archiving in ecology and evolution: how well are we doing? PLoS Biol 13: e1002295
- Rossi T, Nagelkerken I, Simpson SD, Pistevos JC, Watson S-A, Merillet L, Fraser P, Munday PL, Connell SD (2015) Ocean acidification boosts larval fish development but reduces the window of opportunity for successful settlement. Proc R Soc B 282:20151954
- Rummer JL, Stecyk JAW, Couturier CS, Watson SA, Nilsson GE, Munday PL (2013) Elevated CO₂ enhances aerobic scope of a coral reef fish. Conserv Physiol 1:cot023
- Simpson SD, Munday PL, Wittenrich ML, Manassa R, Dixson DL, Gagliano M, Yan HY (2011) Ocean acidification erodes crucial auditory behaviour in a marine fish. Biol Lett 7:917–920
- Sundin J, Jutfelt F (2016) 9–28 d of exposure to elevated *p*CO₂ reduces avoidance of predator odour but had no effect on behavioural lateralization or swimming activity in a temperate wrasse (*Ctenolabrus rupestris*). ICES J Mar Sci 73:620–632
- Tix JA, Hasler CT, Sullivan C, Jeffrey JD, Suski CD (2017) Elevated carbon dioxide has the potential to impact alarm cue responses in some freshwater fishes. Aquat Ecol 51:59–72
- Welch MJ, Watson S-A, Welsh JQ, McCormick MI, Munday PL (2014) Effects of elevated CO₂ on fish behaviour undiminished by transgenerational acclimation. Nature Clim Change 4:1086– 1089